\begin{table}%t1
\par
\caption{\label{TAB-targets}AGB star candidates and targets observed. } 
\par
%\centerline
 {\tiny\begin{tabular}{rlrrrrrl} \hline\hline
2MASS identifier & Other  & $J$ & $H$ & $K$ & $M_{\rm bol}$ & Spectral & Remarks; (references) \\
 & identifier &  & & &  & type &  \\
\hline
2MASS02380397-3420196  & Fornax1 & 15.764 & 14.836 & 14.499 & --3.24 & K3-K4: & \\ 
2MASS02401824-3428412  & Fornax2 & 15.914 & 15.136 & 14.542 & --3.10 & G8-K4: & \\
2MASS02401650-3415374  & Fornax3 & 16.122 & 15.137 & 14.853 & --2.88 & G8-K4: & \\
2MASS02414248-3421284  & Fornax4 & 15.797 & 14.912 & 14.485 & --3.21 & K2-K3 & \\  % K2 better when data is dereddened with Av=1
2MASS02400772-3425156  & Fornax5 & 16.195 & 15.460 & 14.845 & & -- & not observed  \\
2MASS02394347-3413379  & Fornax6 & 15.981 & 15.024 & 14.292 & & -- & not a point source, no spectrum taken \\
2MASS02401171-3431187  & Fornax7 & 16.027 & 15.131 & 14.658 & --2.99 & K2-K3 & \\
2MASS02403857-3428344  & Fornax8 & 15.789 & 14.813 & 14.551 & --3.21 & K3-K4: & \\
2MASS02405606-3431273  & Fornax9 & 15.917 & 15.064 & 14.541 & & -- & not observed  \\
2MASS02395732-3431211  & Fornax10 & 15.907 & 15.095 & 14.666 & & -- & not observed \\
2MASS02405333-3412130  & Fornax11 & 15.034 & 13.981 & 13.261 & --4.01 & C & 1.53 absent; (7) \\ %SR 230
2MASS02402188-3406309  & Fornax12 & 15.729 & 14.843 & 14.474 & --3.27 & K3-K4 & \\
2MASS02400666-3423222  & Fornax13 & 14.485 & 13.377 & 12.618 & --4.63 & C & 1.53 weak; (1, 2, 7) \\ %variable w/o period
2MASS02405288-3428303  & Fornax14 & 15.706 & 14.784 & 14.343 & --3.31 & K2-K3 &\\
2MASS02400946-3406256  & Fornax15 & 15.790 & 14.556 & 13.668 & --3.56 & C & 1.53 strong\\
2MASS02401349-3426192  & Fornax16 & 15.930 & 14.794 & 14.652 & --3.07 & not C & H band not usable  \\
2MASS02403123-3428441  & Fornax17 & 14.745 & 13.689 & 13.072 & --4.23 & C & 1.53 weak; (3, 5, 7) \\ %variable w/o per
2MASS02390184-3427528  & Fornax18 & 15.672 & 14.987 & 14.302 & --    & non stellar & 2MASX source \\
2MASS02405840-3408022  & Fornax19 & 15.994 & 15.030 & 14.591 & --3.03 & K2-K4 & \\
2MASS02401207-3426255  & Fornax20 & 15.131 & 13.732 & 12.728 & --4.53 & C & 1.53 weak; (7) \\ % SR 375
2MASS02401778-3427357  & Fornax21 & 15.424 & 14.122 & 13.182 & --4.05 & C & 1.53 weak; (7) \\ % SR 340
2MASS02400252-3403272  & Fornax22 & 15.855 & 15.157 & 14.608 & --3.15 &  K1-K2 & \\
2MASS02404718-3438510  & Fornax23 & 15.602 & 14.911 & 14.199 & --     & non stellar & 2MASX source \\
2MASS02395421-3438368  & Fornax24 & 15.601 & 14.162 & 13.167 & --4.09 & C & 1.53 weak; (7) \\ % variable w/o period
2MASS02391232-3432450  & Fornax25 & 14.722 & 13.262 & 12.120 & --5.18 & C & very red, features weak, dust emission?; (5,6,7) \\%mira 470
2MASS02394528-3431581  & Fornax26 & 15.941 & 14.989 & 14.667 & & -- & not observed \\
2MASS02410355-3448053  & Fornax27 & 14.441 & 13.365 & 12.694 & --4.59 & C & 1.53 absent; (3,4,7) \\ %mira 280
2MASS02402039-3437345  & Fornax28 & 15.930 & 14.875 & 14.649 & & -- & not observed \\
2MASS02402554-3441585  & Fornax29 & 15.784 & 14.926 & 14.439 & & -- & not observed \\
2MASS02410145-3437154  & Fornax30 & 15.803 & 14.737 & 14.401 & & -- & not observed \\
2MASS02380618-3431194  & Fornax31 & 16.052 & 14.483 & 13.315 & --4.03 & C & 1.53 medium\\
2MASS02385700-3446340  & Fornax32 & 14.789 & 13.664 & 13.076 & --4.21 & C & 1.53 absent; (7) \\ %variable w/o period
2MASS02392952-3449431  & Fornax33 & 15.905 & 15.144 & 14.644 & & -- & not observed \\
2MASS02385056-3440319  & Fornax34 & 16.106 & 14.525 & 12.879 & --4.70 & C & 1.53 extreme; 
(5, 6, 7) \\ % mira 350
                       & Fornax-S66  & 14.768 & 13.901 & 13.642 & & M2S & \# 66  from Stetson et~al. (\cite{Stetson98}) \\
                       & Fornax-S71  & --      & --      & --      & & K5  & \# 71  from Stetson et~al. (\cite{Stetson98}) \\
                       & Fornax-S99  & --      & --      & --      & & S   & \# 99  from Stetson et~al. (\cite{Stetson98}) \\
                       & Fornax-S116 & 15.004 & 14.041 & 14.028 & & SC  & \# 116 from Stetson et~al. (\cite{Stetson98}) \\2MASS01002467-3352012  & Scl1          & 15.024 & 14.280 & 13.791 & --2.92 & non stellar & 2MASX source \\
2MASS01005598-3352344  & Scl2          & 15.977 & 15.290 & 14.610 & --1.97 & K2-K4 & $H$~band curvature odd (background?) \\
2MASS00590481-3343297  & Scl3          & 16.118 & 15.283 & 14.688 & & -- & not a point source, no spectrum taken \\
2MASS01004469-3327134  & Scl4          & 15.367 & 14.643 & 14.133 & --2.57 & non stellar & 2MASX, 2dFGRS S503Z077 ($z=0.11$) \\
2MASS00595925-3337091  & Scl5          & 15.963 & 15.284 & 14.565 & & -- & not observed \\
2MASS00595367-3338308  & Scl6          & 14.846 & 13.144 & 11.603 & --4.93 & C & 1.53 strong; (4) \\
2MASS00593953-3332545  & Scl7          & 15.264 & 14.501 & 14.018 & --2.67 & non stellar & 2MASX source \\
2MASS01005213-3343136  & Scl8          & 15.845 & 15.171 & 14.543 & --2.10 & non stellar & 2MASX source \\                       & Scl-Az1-C     & 14.713 & 14.040 & 13.871 & & C  & ALW1 from Azzopardi et~al. (\cite{Azzopardi85}) \\ %J00583615-3340258  
                       & Scl-Az3-C     & --      & --      & --      & & C  & ALW3 from Azzopardi et~al. (\cite{Azzopardi85}) \\ %00 59 58.9 -33 28 34 
                       & Scl-scms-209  & 13.628 & 13.080 & 12.823 & & M4 & from Schweitzer et~al. (\cite{Schweitzer95}) \\ %J00595802-3345174  
                       & Scl-scms-1448 & 15.167 & 14.497 & 14.340 & & M3 & from Schweitzer et~al. (\cite{Schweitzer95}) \\ %J00594404-3341158
                       &      \\
2MASS19445931-1446010  & N6822-1  & 16.222 & 15.148 & 14.737 & & -- & not observed  \\
2MASS19444196-1440072  & N6822-2  & 15.889 & 14.766 & 14.417 & --5.93 & K4-M & \\
2MASS19450019-1446040  & N6822-3  & 15.189 & 14.234 & 13.760 &  & -- & not observed \\
2MASS19443399-1446237  & N6822-4  & 15.593 & 14.519 & 14.231 & --6.22  & M &  H$_2$O absorption\\
2MASS19445849-1448037  & N6822-5  & 16.142 & 15.314 & 14.785 & & -- & double star, no spectrum taken \\
2MASS19444529-1451252  & N6822-6  & 16.083 & 15.059 & 14.553 & & -- & not observed \\
2MASS19445689-1442381  & N6822-7  & 16.064 & 15.104 & 14.610 & --5.76 & M & H$_2$O absorption \\
2MASS19444914-1449051  & N6822-8  & 16.042 & 14.913 & 14.633 & --5.78 & K4-M & \\
2MASS19444900-1444334  & N6822-9  & 15.230 & 14.204 & 13.738 & & -- & not observed \\
2MASS19452775-1453299  & N6822-10 & 15.383 & 14.415 & 14.022 & --6.43 & M & H$_2$O absorption\\
                       &      \\
                       & LHS 517   &  6.510 &  5.899 &  5.594 & & M3.5 dwarf &  \\ %redflag=111
                       & LHS 3788B &  9.459 &  8.840 &  8.531 & & M5 dwarf &  \\
\hline
\end{tabular}}
\par
\smallskip
1 = Demers \& Kunkel (\cite{Demers79});
2 = Westerlund et~al. (\cite{Westerlund87});
3 = Bersier \& Wood (\cite{Bersier02});
4 = Mauron et~al. (\cite{Mauron04}); 
5 = Matsuura et~al. (\cite{Matsuura07});  
6~=~Lagadec et~al. (\cite{Lagadec08}); 
7 = Whitelock et~al. (\cite{Whitelock09}).
\par
\end{table}